Author's response to reviews

Title: The relationship between glucocorticoid receptor polymorphisms, stressful life events, social support, and post-traumatic stress disorder

Authors:

Yulong Lian (lianyulong444@163.com)
Jing Xiao (xiaojing198088@163.com)
Qian Wang (qqian@xjmu.edu.cn)
Li Ning (ninglili@xjmu.edu.cn)
Suzhen Guan (gszhen@xjmu.edu.cn)
Hua Ge (GGHUA@xjmu.edu.cn)
Fuye Li (lfuye129@xjmu.edu.cn)
Jiwen Liu (liujiwendr@xjmu.edu.cn)

Version: 2 Date: 28 March 2014

Author's response to reviews:

Dear Editors:
Thank you for your revision, we had revised our manuscript to include line and page numbers and competing interests.
Best Regards,
Yulong Lian Ph.D.
Division of Occupational and Environmental Health
College of Public Health
Nantong University
Email: lianyulong444@163.com